Author's response to reviews

Title: A community-based cluster randomised trial of safe storage to reduce pesticide self-poisoning in rural Sri Lanka: study protocol

Authors:

Melissa Pearson (melissa@sactrc.org)
Flemming Konradsen (flko@sund.ku.dk)
David Gunnell (D.J.Gunnell@bristol.ac.uk)
Andrew H Dawson (adawson@sactrc.org)
Ravi Peiris (ravi@sactrc.org)
Manjula Weerasinghe (manjugaya@yahoo.com)
Duleeka W Knipe (duleekaknipe@yahoo.co.uk)
Shaluka Jayamanne (shalukajaya@yahoo.com)
Chris Metcalfe (Chris.Metcalfe@bristol.ac.uk)
Keith Hawton (keith.hawton@psych.ox.ac.uk)
A Rajitha Wickramasinge (arwicks@slt.net.lk)
W Atapattu (pdhsncplk@gmail.com)
Palitha Bandara (palithabandara120@yahoo.com)
Dhammika de Silva (pmuncplk@gmail.com)
Asanga Ranasinghe (venurarana@gmail.com)
Fahim Mohamed (fahim@sactrc.org)
Nicholas A Buckley (n.buckley@unsw.edu.au)
Indika Gawarammana (indika@sactrc.org)
Michael Eddleston (M.Eddleston@ed.ac.uk)

Version: 2 Date: 11 October 2011

Author's response to reviews:

Dear Editors

I have reformatted the paper to move the trial registration number to the end of the abstract section as requested. I also made the changes in the main document.